Author's response to reviews

Title: Early Life Environment and Natural History of Inflammatory Bowel Diseases

Authors:

Abra Guo (aguo1@mgh.harvard.edu)
Betsy Stevens (bstevens1@partners.org)
Robin Wilson (rgwilson@mgh.harvard.edu)
Caitlin Russell (cnrussell@mgh.harvard.edu)
Melissa Cohen (mcohen4@mgh.harvard.edu)
Holly Sturgeon (hsturgeon@mgh.harvard.edu)
Anna Thornton (athornton4@mgh.harvard.edu)
Cosmas Giallourakis (cgiallourakis@mgh.harvard.edu)
Hamed Khalili (hkhalili@mgh.harvard.edu)
Deanna Nguyen (deanna.nguyen@mgh.harvard.edu)
Jenny Sauk (jsauk@partners.org)
Vijay Yajnik (vyajnik@partners.org)
Ramnik J Xavier (rxavier@mgh.harvard.edu)
Ashwin N Ananthakrishnan (aananthakrishnan@partners.org)

Version: 7  Date: 1 December 2014

Author’s response to reviews:

I have provided both financial and non-financial competing interests.